

The Mars 2020 Candidate Landing Sites: A Magnetic Field Perspective

Anna Mittelholz, Achim Morschhauser, Catherine Johnson, Benoît Langlais, Robert Lillis, Foteini Vervelidou, Benjamin Weiss

▶ To cite this version:

Anna Mittelholz, Achim Morschhauser, Catherine Johnson, Benoît Langlais, Robert Lillis, et al.. The Mars 2020 Candidate Landing Sites: A Magnetic Field Perspective. Earth and Space Science, 2018, 5 (9), pp.410-424. 10.1029/2018EA000420. hal-02176327

HAL Id: hal-02176327 https://hal.science/hal-02176327

Submitted on 8 Jul 2021

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers.

L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution - NonCommercial - NoDerivatives 4.0 International License



Earth and Space Science



10.1029/2018EA000420

Key Points:

- Mars 2020 offers the opportunity to acquire samples that record the intensity and direction of the ancient Martian magnetic field
- Laboratory paleomagnetic measurements of returned samples can address questions about the history of the Martian dynamo, thermal evolution, and climate
- We recommend Northeast Syrtis as preferred site for magnetic investigations, followed by Columbia Hills and Jezero

Correspondence to:

A. Mittelholz, amittelh@eoas.ubc.ca

Citation:

Mittelholz, A., Morschhauser, A., Johnson, C. L., Langlais, B., Lillis, R. J., Vervelidou, F., & Weiss, B. P. (2018). The Mars 2020 candidate landing sites: A magnetic field perspective. *Earth and Space Science*, *5*, 410–424. https://doi.org/10.1029/2018EA000420

Received 5 JUN 2018 Accepted 24 JUL 2018 Accepted article online 30 JUL 2018 Published online 12 SEP 2018

©2018. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

The Mars 2020 Candidate Landing Sites: A Magnetic Field Perspective

Anna Mittelholz¹, Achim Morschhauser², Catherine L. Johnson^{1,3}, Benoit Langlais⁴, Robert J. Lillis⁵, Foteini Vervelidou², and Benjamin P. Weiss⁶

¹Department of Earth, Ocean and Atmospheric Sciences, The University of British Columbia, Vancouver, British Columbia, Canada, ²GFZ German Research Center for Geosciences, Potsdam, Germany, ³Planetary Science Institute, Tucson, AZ, USA, ⁴Laboratoire de Planétologie et Geodynamique, UMR 6112 CNRS & Université de Nantes, Nantes, France, ⁵Space Science Laboratory, University of California, Berkeley, CA, USA, ⁶Department of Earth, Atmospheric, and Planetary Sciences, Massachusetts Institute of Technology, Cambridge, MA, USA

Abstract We present an analysis of the remaining three candidate landing sites for Mars 2020, Columbia Hills (CH), Northeast Syrtis (NES) and Jezero (JE) from the perspective of understanding Mars' crustal magnetic field. We identify how the different sites can address each of six community-defined paleomagnetic science objectives for Mars return samples. These objectives include understanding the early dynamo field and its variability, identification of magnetic minerals that carry magnetization in the samples, and characterization of any thermal and chemical alteration of samples. Satellite data have provided global and regional constraints on crustal magnetization, indicating strong magnetizations at CH and weak to no magnetization at JE and NES. However, the primary paleomagnetic interest—understanding the early dynamo—requires ground truth from a landing site at which pre-Noachian and Early Noachian deposits are accessible. This requirement is most likely met by the site NES, which contains meggabreccia deposits, and it is therefore the highest priority landing site for magnetic field investigations. Importantly, a sample return mission has never been done, and so any of the three landing sites will provide critical, new data that will contribute to understanding the history of Mars' magnetic field and crustal mineralogy and, in turn, yield constraints on the planet's evolution.

Plain Language Summary Mars 2020, a mission to Mars, will for the first time collect rock samples for subsequent return to the Earth. Here we evaluate how each of the last three candidate landing sites can shed light on the Martian magnetic field and its history. We evaluate which landing site is best suited to address each of six science objectives regarding the origin, variability, and evolution of the magnetic signatures on Mars. We first summarize what is known about magnetization around the landing sites based on satellite data. Because we are primarily interested in the history of the magnetic field, an ideal site should offer ancient (~4 billion years old and older) accessible outcrops that might be recorders of Mars' ancient global magnetic field. We combine current knowledge about the surface age, geology, and magnetic field, and chose Northeast Syrtis as the preferred landing site, followed by Columbia Hills and Jezero. However, any of the three sites will provide us with new data that will contribute to understanding Mars' magnetic field history and, in turn, its evolution.

1. Introduction

The Martian magnetic field is unique among the planets in our solar system. Satellite magnetometry measurements at Mars in the 1990s revealed strong crustal fields (Acuna et al., 1999). Their sources are magnetic minerals in the crust that were likely magnetized by an ancient global dynamo field that was active until about 4 billion years ago (Lillis et al., 2013; Vervelidou, Lesur, Grott, et al., 2017). The strongest crustal fields are concentrated in the southern hemisphere, where they exceed the Earth's lithospheric field strength by an order of magnitude at orbital altitudes (Purucker et al., 2000). Explanations for such strong and localized fields have been widely discussed; strong remanent magnetizations and/or a thick magnetized layer are typically invoked (Connerney et al., 1999; Langlais et al., 2004), but no consensus has been reached as to the underlying magnetization mechanism. Additionally, the origin, strength, and temporal decline of the ancient dynamo field are poorly constrained. Paleomagnetic measurements of samples, selected by the Mars 2020 rover mission and subsequently returned by a downstream mission, could revolutionize our understanding of the ancient dynamo field, that in turn is coupled to the evolution of the planet's interior (Kuang et al., 2008), tectonic history (Stevenson, 2001), and climate (Jakosky & Phillips, 2001).

Magnetic field data have primarily been collected by satellites, particularly the Mars Global Surveyor (MGS) and Mars Atmosphere and Volatile EvolutioN (MAVEN) spacecraft. A further source of magnetic field information is the ancient and strongly magnetized Martian meteorite, ALH 84001, which suggests that a dynamo with an Earth-strength field was active at, or before, 4 Ga (Weiss et al., 2002). Other Martian meteorites, that is, the shergottite, nakhlite, and chassignite meteorites, provide insight into the formation and intrinsic magnetic properties of Martian rocks (Rochette et al., 2005) and indicate that the dynamo had declined to no more than \sim 1 µT at 1.3 Ga (e.g., Collinson, 1997; Funaki et al., 2009). Because these Martian meteorites could have been magnetized by a crustal remanent field rather than an active dynamo, the results only require that a dynamo existed at, or prior to, the timing of magnetization acquisition. The InSight Lander will be the first mission to measure the magnetic signal at the planetary surface (Banerdt et al., 2013). Its measurements will however be limited to the landing location. The upcoming sample return mission Mars 2020 offers a unique chance to sample magnetized rock from the surface of Mars with geological context, sample orientation, and, after the samples are returned, precise radiometric ages.

The mission objectives for Mars 2020 are focused on investigating Mars as potential habitat for past or present life (Mustard et al., 2013). Magnetic studies contribute to such investigations because a global magnetic field can provide conditions favorable for life as we know it (Ehlmann et al., 2016; Rochette et al., 2006; Weiss et al., 2018). Mars has lost most of its atmosphere throughout its history, possibly related to the cessation of its global dynamo field, which may have facilitated atmospheric erosion by the solar wind (Dehant et al., 2007). The cessation of the dynamo is typically cited as ~4.05-4.1 Ga (Lillis et al., 2008, 2013; Vervelidou, Lesur, Grott, et al., 2017), although later/longer-duration dynamos have been proposed (Langlais & Purucker, 2007; Schubert et al., 2000; Milbury et al., 2012). In particular, if the coherence length-scale of the magnetization in younger regions is below the satellite resolution, there may be smaller regions of younger units with primary magnetizations acquired in a field that persisted after 4.1 Ga. Resolution of the magnetic surface field can be approximated by orbit altitude (Blakely, 1995), and the lowest satellite measurements to date are mostly at around 135-km altitude with a few measurements taken at ~100-km altitude. Constraining the properties and the cessation of the dynamo with higher accuracy requires sampling of representative rocks formed during the pre-Noachian to Early Noachian, the period in which the dynamo most likely ceased. This presents a challenge to Mars 2020, as pre-Noachian materials are at best known to be exposed at very limited locations and over small spatial scales on Mars (Nimmo & Tanaka, 2005). Such materials may have undergone substantial thermal and chemical reworking over Mars' history.

Sample return will address long-standing questions regarding the nature and origin of crustal magnetization, which are fundamentally linked to the planet's early history and evolution (Weiss et al., 2018). In particular, the mechanisms responsible for magnetization acquisition and the mineral carriers are key unknowns. Ferroand ferrimagnetic minerals can acquire natural remanent magnetization (NRM) when they are exposed to a magnetic field via thermal, chemical, and/or shock processes. These three mechanisms respectively result from (i) magnetic minerals cooling below their blocking temperature, (ii) alteration and magnetization during (re)crystallization, and (iii) shock during impacts. Magnetization measurements and age dating on actual samples will shed light on chronology of the dynamo field, and/or mechanisms of magnetization acquisition and/or removal. In contrast to meteorite samples, returned and oriented samples will allow the paleodirection of the ancient field to be inferred and provide samples unaffected by the shock associated with planetary ejection of meteorites, although possibly affected by shock associated due to impactors. Thus, laboratory measurements of sample magnetization will provide important constraints on magnetization models. Such models currently rely solely on satellite data that provide little to no information on short wavelength crustal fields. Furthermore, inferring the underlying magnetization from magnetic field measurements is an inherently nonunique inverse problem because only part of the underlying magnetization gives rise to the observable magnetic field (e.g., Gubbins et al., 2011; Maus & Haak, 2003; Runcorn, 1975).

The purpose of this paper is to review existing knowledge regarding the possible Mars 2020 landing sites Columbia Hills (CH), Northeast Syrtis (NES), and Jezero (JE), with a focus on suitability for obtaining samples that can shed light on the linked issues regarding magnetic field history and evolution, and crustal magnetization. In the following, we give a geologic overview and assess the final three candidate landing sites in the context of future magnetic investigation of returned samples. We briefly review recent global and regional models that provide constraints on magnetization and crustal field intensity at each site. We then summarize paleomagnetic sample return objectives and discuss what further knowledge could be gained using returned

Earth and Space Science





samples. We evaluate each of the landing sites in the context of these objectives and rank the landing sites accordingly, noting that any of the three sites will provide important new information on Mars' magnetic field and crustal magnetization.

2. Local Geology

The final three candidates for the Mars 2020 landing site are CH (15° S, 176° E), NES (18° N, 77° E), and JE (18° N, 79° E) as shown in Figure 1. All landing sites are located along the global crustal dichotomy (Smith et al., 2001). NES and JE lie close to the western rim of the Isidis impact basin, and CH is within the Gusev impact basin and close to the former landing site of the Spirit rover. Figures 1c and 1d show the CH landing site location within Gusev crater. The rocks exposed at the surface of CH are thought to be Late Noachian-Early Hesperian volcanoclastic and possibly also evaporitic materials draped on topography and uplifted by the impact that formed Gusev or surrounding craters (McCoy et al., 2008; van Kan Parker et al., 2010). The area offers a variety of sampling possibilities because igneous units, as well as clays and carbonate-rich outcrops have been detected, that may be of hydrothermal (Filiberto & Schwenzer, 2013) or possibly even lacustrine origin (Morris et al., 2010; Ruff et al., 2014). The finding of opaline silica by Spirit, possibly the result of fumarolic sulfate leaching or hot spring silica precipitation (Ruff et al., 2011; Squyres et al., 2008) makes this landing site an interesting candidate.

The two landing sites NES and JE in the Nili Fossae region are not distinguishable on a global scale as they are only 37 km apart (Figure 1a). The JE landing site is within JE crater, and NES is about 15 km southwest of and outside the crater rim (Figures 1e and 1f). The NES landing site contains one of the most complete



Earth and Space Science



Figure 2. Gusev: (a) Histogram showing model magnetization distributions for each geological time period after Tanaka et al. (2014) where the investigated area (red square in c) spans $\pm 5^{\circ}$ of the approximate location of the landing site. (b) Model magnetic field magnitude, |B|, at the planetary surface. (c) Geologic map (Tanaka et al., 2014) AHi = Amazonian and Hesperian impact unit; AHtu = Amazonian Hesperian transition undivided unit; eHv = Early Hesperian volcanic unit; eNh = Early Noachian highland unit; Hve = Hesperian volcanic edifice unit; HNt = Hesperian Noachian transition unit; INH = Late Noachian highland unit; mNH = Middle Noachian highland unit. (d) Model magnetization, |M|

sequences of ancient (pre-Noachian to Early Hesperian) igneous and possibly also sedimentary rocks on Mars. A phyllosilicate-rich igneous basement unit is overlain by olivine-carbonate rocks and a mafic capping unit (Bramble et al., 2017). An extended Mars 2020 mission might also access sulfate-rich rocks and Hesperian lavas. JE crater contains clastic sediments deposited as part of a fluvio-deltaic-lacustrine system during the Late Noachian to Early Hesperian, possibly covered in some locations by younger igneous rocks. The sediments are sourced from pre-Noachian to Early Hesperian NES regional rocks, similar to those exposed at the NES landing site. An extended mission outside the crater might be able to access these ancient rocks directly. Hence, sampling deltaic and lacustrine sediments of a paleolake is possible. Smectites and carbonate-bearing basin fill may be the result of aqueous alteration of olivine-rich material. Younger volcanic areas (Early Amazonian) complement the variety of possible samples reachable from the JE landing site (Ehlmann & Mustard, 2012; Hiesinger & Head, 2004). Outside the crater and closer to NES, olivine-carbonate rocks have been observed that are serpentinized and indicative of aqueous alteration. Thus, samples from acid sulfate formations, layered phyllosilicates, and volcanic and plutonic material are possible.



Figure 3. Jezero (black star) and NE Syrtis (red star): (a) Histogram showing model magnetization distributions for each geological time period after Tanaka et al. (2014), where the investigated area (red square in c) spans $\pm 5^{\circ}$ of the approximate location of the landing site. (b) Model magnetic field magnitude, |B|, at the planetary surface. (c) Geologic map (Tanaka et al., 2014). AHi = Amazonian and Hesperian impact unit; eHv = Early Hesperian volcanic unit; eNh = Early Noachian highland unit; HNt = Hesperian Noachian transition unit; IHI = Late Hesperian lowland unit; IHt = Late Heperian transition unit; mNh = Middle Noachian highland unit; mNhm = Middle Noachian highland massif unit. (d) Model magnetization, |M|

3. Magnetization Models

3.1. Methodology

Several magnetic field and magnetization models have been developed using MAVEN and MGS data and allow representations of magnetization and the magnetic field at different altitudes down to the planetary surface (e.g., Arkani-Hamed, 2004; Cain, 2003; Langlais et al., 2004, 2017; Mittelholz et al., 2017; Morschhauser et al., 2014; Purucker et al., 2000). Independent of the modeling approach, a major challenge is the nonuniqueness of such inversions and the resulting suites of possible models that can explain the observations equally well. A way of dealing with this is to assume a priori information. One such approach is to impose constraints on the geometry of the magnetized structure (e.g., Hood & Zakharian, 2001; Takahashi et al., 2014). Another involves solving the inverse problem for the minimum magnetization required to explain the magnetic field observations (Parker, 2003; Whaler & Langel, 1996; Whaler & Purucker, 2005). In this study we use two different approaches to model magnetization at the three landing sites using spacecraft data: a global and a regional approach.

Our global approach to estimate magnetization follows the work of Vervelidou, Lesur, Morschhauser, et al. (2017). In this approach, the magnetization was constrained by decomposing it into three parts, each being the projection on one of the three vector spherical harmonic (SH) basis functions as proposed by Gubbins et al. (2011). Of these, only one gives rise to the observable magnetic field; the other two span the null space of the inversion. Inverting a given SH magnetic field model only for the former, called the visible magnetization, Vervelidou, Lesur, Morschhauser, et al. (2017) yielded a solution that fits the model perfectly and is unique for a magnetized layer of given thickness, over which the magnetization varies only laterally. Due to the uniqueness of the solution, the inverse problem reduces to a simple conversion formula between the SH coefficients of the

Table 1

Key Paleomagnetic Science Objectives for a Returned Mars Sample

	a mais campie			
Science Objectives	Site Requirements	NES	СН	JE
1. Determine the intensity of the Martian	Samples old enough (pre-Noachian and Early	***	**	**
dynamo.	Noachian) to have likely been magnetized in			
	ancient dynamo magnetic field.			
2. Characterize the dynamo reversal frequency	Samples should span a dateable large time	***	**	**
and conduct magnetostratigraphy.	interval, $\sim>$ 1 Ma, during Early Noachian;			
	orientation should be known to within 30°.			
3. Constrain the effects of (i) heating and (ii)	Variety of samples: (i) heated samples	**	***	**
aqueous alteration on the samples.	and (ii) evidence of water at the surface.			
4. Test the hypotheses that Mars experienced	Parent rock of sample should be in-place	***	**	*
plate tectonics and/or true polar wander and	bedrock or at least contain paleohorizontal			
constrain the tectonic and deformational	indicators such as bedding planes or stratified			
history of the landing site.	grain size sorting.			
5. Determine the major mineral carriers of	Site should offer a variety of mineralogies.	**	***	**
Martian crustal magnetization.				
6. Constrain sediment sourcing, fluid flow,	Site should offer environments where evidence	**	***	**
and the depositional environment using	for sediment deposition and fluid flow			
environmental magnetism studies.	exists.			

Note. Key paleomagnetic science objectives for a returned Mars sample adapted from Weiss et al. (2018) with their respective requirements and a rating of landing sites, where three asterisks mean the highest chance and one asterisk means the lowest chance to fulfill requirements for Northeast Syrtis (NES), Columbia Hills (CH), and Jezero (JE).

magnetic field and the vector SH coefficients of the visible magnetization. Based on this method, Vervelidou, Lesur, Morschhauser, et al. (2017) derived a visible magnetization model of Mars, assuming a magnetized layer of 40 km and using the SH model of Morschhauser et al. (2014), derived from MGS data. The magnetization intensity is shown in Figure 1b.

We also solve for the magnetization (Figures 2d and 3d) required to explain magnetic field observations using equivalent source dipoles as introduced by Mayhew (1979). The observed measurements at any point can be described by the sum of the fields due to regularly spaced dipoles (here every 115 km) placed within a constant thickness layer (here 40 km). The input data set is generated by subselecting MGS and quiet MAVEN data, as described in Mittelholz et al. (2018). For a set of possible model solutions, we examine the L-curve to find a reasonable trade-off between model complexity and misfit (Aster et al., 2011) and use the resulting model to predict the surface field (Figures 2b and 3b). Here we show regional models of the magnetization and the magnetic field predicted at the surface around the landing sites (Figures 2b, 2d and 3b, 3d).

3.2. Global Context

Measured from orbit, the strong crustal magnetic fields are mostly confined to the southern hemisphere, where they exceed the Earth's lithospheric field strength by an order of magnitude, and show substantially weaker but nonetheless significant signatures in the north. Magnetization as shown in Figure 1b is broadly correlated with topographically higher and on average older areas (Figure 1a; Tanaka et al., 2014). Both exogenic (impact-related differences in crustal structure) and endogenic (hemispheric dynamo and/or internally driven differences in the distributions of magnetic carriers) processes have been invoked to account for the north-south magnetic dichotomy (Amit et al., 2011; Andrews-Hanna et al., 2008; Marinova et al., 2008; Monteux et al., 2015; Quesnel et al., 2009; Stanley et al., 2008; Watters et al., 2007). The JE and NES landing sites are located west of Isidis, the youngest of the large (>1,000 km) unmagnetized impact basins. The lack of magnetization suggests that the JE impact occurred after the shutdown of the global dynamo and either impacted partially demagnetized crust or resulted in demagnetization without new acquisition of magnetization. CH is located in an area of appreciable magnetization north of the strongest magnetic fields on Mars.

3.3. Regional Models

A more detailed picture of the magnetic environment of the candidate sites, derived from regional magnetization models, is shown in Figures 2 and 3. The predicted magnetic field at the surface of the CH landing site reaches around 400 nT with a magnetization of 1 A/m, under the assumption of a 40-km magnetized layer. The mean values across the entire modeled region are 900 nT and 2 A/m. Predicted model magnetization and surface fields at the JE and NES landing sites are around 0.1 A/m and 35 nT, respectively, with mean values of 0.2 A/m and 100 nT across the entire modeled area in Figure 3. Values below 0.2 A/m and 100 nT (shaded gray in Figures 2b, 2d, 3b, and 3d) may however be considered below noise levels and are thus consistent with no magnetization and no magnetic field, respectively. However, localized fields could be much stronger, as only wavelengths of around 130 km can be resolved by currently available satellite data from Mars.

4. Paleomagnetic Science Objectives and Candidate Landing Site Assessment

A community-based study (Weiss et al., 2018) developed a ranked list of science objectives (column 1 in Table 1) for paleomagnetic studies with returned samples. In the following, we discuss each of these science objectives, and the resulting site requirements (column 2), and evaluate the potential of the proposed Mars 2020 landing sites to address these (column 3).

4.1. SO1 — Paleodynamo Intensity

4.1.1. Science Objective

The paleointensity of the Martian core dynamo field can be constrained if the magnetization and ferromagnetic mineral content of the crust is known. Such information cannot be recovered from satellite data because of nonuniqueness (Gubbins et al., 2011; Runcorn, 1975) and because the magnetic mineral content of the deeper crust cannot be inferred from orbital spectroscopic measurements. Therefore, the only direct constraint on the intensity of the Martian dynamo magnetic field is based on the meteorite ALH 84001, which contains 4.1-Ga-old thermoremanence and suggests a paleofield within the same order of magnitude as the present Earth main field, i.e. \sim 50 µT (Gattacceca & Rochette, 2004; Weiss et al., 2008). However, the inferred magnetization is too weak to explain the locally strong magnetic field anomalies associated with some of the old Noachian terrains. The meteorite has a complex postformational history with multiple thermal and deformational events (including ejection from Mars) that may have altered its original magnetization (Weiss et al., 2008). The need for samples magnetized during an active core dynamo is therefore the first paleomagnetic science objective for Mars.

This objective requires a site where material magnetized by a paleodynamo field is easily accessible. Based on inferences regarding the dynamo history from satellite data, this implies that a candidate site should include pre-Noachian and Early Noachian terrain.

4.1.2. Site Evaluation

To address SO1, we first evaluate the magnetization strength in the regions around the candidate landing sites CH (red square in Figure 2c) and JE/NES (red square in Figure 3c), by assigning each $0.5^{\circ} \times 0.5^{\circ}$ region (~900 km²) its mean magnetization value and the dominant stratigraphic period. At this scale, the strongest and most spatially extensive magnetic anomalies are associated with Noachian terrain at CH (Figure 2a), whereas most of the predicted magnetization for JE and NES is very weak, below 0.6 A/m, and is associated with Hesperian terrain. From this analysis, the likelihood of picking up a magnetized rock of Noachian age appears to be low at the JE and NES landing sites. However, given that the rocks exposed at the JE and CH landing sites are thought to be no older than Late Noachian, they are unlikely to provide bulk rocks that are magnetized if the dynamo was active only during the pre-Noachian to Early Noachian. In contrast, the megabreccia basement at NES, thought to have been produced by the Isidis impact, and therefore Middle to pre-Noachian in age (Fassett & Head, 2011; Frey, 2008; Werner, 2008) is a high-priority target for paleomagnetic studies. Megabreccia, a cataclastic deposit of large (\geq 10-m diameter) blocks within a finer-grained matrix, may represent the oldest exposed and accessible material. If layered, they provide a window into intact stratigraphy of ancient crust (Weiss et al., 2018). For the JE landing site, although the rocks are young, some mineral grains and clasts in the sediments could be as old as pre-Noachian because they are derived from the JE watershed that includes megabreccia. An extended mission could actually access the old JE watershed rocks directly. Gusev crater is the catchment area of Ma'adim Vallis, which is carved in Early and Middle Noachian material and could also contain material from those units. Furthermore, the magnetic field at NES and JE may be weak because the Isidis impact disrupted the magnetized crust, thereby removing its spatial coherence and consequently its magnetic signature at orbital altitudes. Substantial local exposures could still be magnetized while not visible magnetically at orbital altitudes. Hydrocode simulations formerly used for the Orientale Basin on the Moon (Johnson et al., 2016) suggest that megabreccia deposits have not been completely demagnetized by the impact. In contrast, Late Noachian and younger magnetized material could be a result of an older underlying magnetized layer, i.e. magnetization due to crustal rather than dynamo fields.

SO1 can therefore best be addressed by sampling layered megabreccia at NES. JE and CH are alternative second choices, because of weak magnetization and possible lack of ancient material, respectively.

4.2. SO2—Paleodynamo Reversals and Magnetostratigraphy 4.2.1. Science Objective

Paleodirection of the ancient field is also of interest. It is known that the terrestrial core dynamo reverses its polarity about every 10⁵ – 10⁶ years (Cox et al., 1964; Merrill & McElhinny, 1983), with intermittent longer periods of stability. Such reversals likely also occurred on Mars, based on our current understanding of dynamo theory and satellite magnetic field data (Connerney et al., 1999; Thomas et al., 2018). However, in most cases it is not possible to associate an age with a given pole orientation, as most crustal anomalies cannot be clearly associated with a dateable surface expression, such as a volcano or impact crater, and more generally because the surface age may not reflect the timing of subsurface magnetization acquisition. On Earth, field reversals are identified from paleomagnetic directions in samples with known georeferenced orientation (Butler, 1992). A definite identification of a reversed polarity would be the first direct evidence for field reversals on Mars or on any other planet besides Earth. Paleomagnetic samples from Mars would provide the first direct measurements of paleofield direction, hopefully with a known absolute age obtained from radiogenic dating. They might eventually even offer the possibility to establish a partial chronology of field reversals for Mars, if enough samples of different ages are available. The frequency of reversals is an important constraint on the dynamics of the core (Driscoll & Olson, 2009; Ryan & Sarson, 2007) and convective regime of the mantle (Olson et al., 2013; Pétrélis et al., 2011). Polarity changes have also been hypothesized to lead to weaker magnetization for prolonged cooling rates and could be an explanation for a lack of magnetization in some Noachian-aged areas (Rochette, 2006).

The identification of distinct magnetic polarity has modest requirements on the precision of paleodirection, that is, knowing the direction to about 30° would be sufficient. The investigation of magnetostratigraphy requires analysis of multiple samples covering the time frame over which the dynamo was active and for a long enough duration to capture reversals. In this respect, the site age requirements of S01 apply to S02, but in addition reconstruction of the original orientation and sample context are required. The paleomagnetic measurements should be coupled with radiometric dating of the samples.

4.2.2. Site Evaluation

To achieve SO2, we need rocks magnetized in a dynamo field where reorientation, by, for example, impacts or tectonics, of the blocks has not occurred, is negligible or traceable. The basement rocks of JE and NES are located close to the demagnetized Isidis basin and may have been reoriented during the impact. Megabreccia in NES represent outcrops spanning an extended contiguous time frame. Some megabreccia exposures near, but not in, the landing site are layered (Weiss et al., 2018), and because different blocks are derived from similar locations in the original bedrock, they possibly provide access up to hundreds of meters of intact stratigraphy. Following the argument in SO1, we argue that megabreccia and therefore NES is the preferred sampling site for SO2. An extended mission would allow access to megabreccia from JE. However, the location close to the Isidis basin could be a disadvantage for NES and JE.

4.3. SO3—Heating, Alteration, and Radiolysis 4.3.1. Science Objective

Because the interest in paleomagnetic samples is mostly focused on ancient samples, thermal or chemical alteration is a key issue in sample selection and analysis. Although the main Mars 2020 mission goals address biosignatures and habitability, alteration or preservation state of the sample also has implications for the magnetic field record. An original magnetization can be altered, overprinted, or entirely erased by subsequent thermoremanent magnetization (TRM) or chemical remanent magnetization (Butler, 1992). Therefore, samples that were unaltered following magnetization acquisition in a core dynamo field are preferred. Vice versa, the acquisition history of TRM in a Martian rock sample may be revealed in the laboratory and the original magnetization and subsequent overprints identified. Rock magnetic studies, together with petrological studies of composition and fabric could identify, for example, magnetic minerals formed during serpentinization, as well as signatures of metamorphism. Collectively, these studies will better inform the roles of aqueous alteration and metamorphism in the magnetization record of the Martian crust (Gunnlaugsson et al., 2006; Harrison & Grimm, 2002; Quesnel et al., 2009) aiding the interpretation of satellite data and better constraining the timing of the core dynamo (Langlais & Purucker, 2007; Lillis et al., 2006).

Different geologic units are more prone to different alteration processes. For example, thermal alteration of paleomagnetic samples is most likely for volcanic units, and chemical alteration involving water may be found in deltaic sediments. Ideally, we are interested in magnetized samples of a variety of host rocks.

4.3.2. Site Evaluation

All three candidate landing sites offer a variety of possibilities for collecting samples from different units, and drills and abrasion tools will help obtain samples minimally affected by surface alteration. Because the origin of the magnetization is not important for SO3, preference would be given to CH, as we expect more widespread magnetization of samples from a variety of sources, for example, volcanic Ma'adim watershed, etc. NES would offer megabreccia but possibly not a broad range of magnetized samples, and magnetized samples around JE could generally be very limited.

4.4. SO4—Plate Tectonics and True Polar Wander 4.4.1. Science Objective

On Earth, estimates of paleopole locations from paleodirections allow to investigate paleosecular variation, changes in the planet's rotation axis (true polar wander), or movement of tectonic plates (apparent polar wander). Paleopole locations are used to reconstruct plate motions (e.g., Metelkin et al., 2010; Weil et al., 1998); however, it is unknown if plate tectonics operated on early Mars (Connerney et al., 2005; Fairén et al., 2002; McKenzie, 1999; Sleep, 1994; Sprenke & Baker, 2000). Also, knowing the magnetization direction at one location may help to further constrain the behavior of the paleodynamo. On Earth, the geocentric axial dipole hypothesis is commonly used to relate location latitude and direction (inclination) of the paleofield. Although we acknowledge that samples from one specific location on Mars would not suffice to distinguish among paleosecular variation, true and apparent polar wander, studying paleopoles on Mars, their current location with respect to the rotation axis, and putting this in a global context will also shed light on the dynamo processes. Addressing this objective will thus help tackle some of the most fundamental questions related to Martian history.

Paleopole locations for Mars have been estimated from satellite magnetic field data by analyzing mostly isolated crustal anomalies (e.g., Hood & Zakharian, 2001; Langlais & Purucker, 2007; Milbury et al., 2012; Thomas et al., 2018). However, large uncertainties are associated with inferring paleopole positions from satellite data (Thomas et al., 2018; Vervelidou, Lesur, Morschhauser, et al., 2017). Paleomagnetic samples from Mars, with known orientation, will help in narrowing down these uncertainties. Similar to SO2, in order to estimate paleodirections from a sample, the geologic context has to provide in-place bedrock or paleohorizontal indicators, but with higher accuracy as for SO2. The paleolatitude of the pole could then possibly be inferred. This objective requires careful diagnosis of outcrops with camera images, where outcrops must contain paleohorizontal indicators, such as bedding planes, flow top features, vesicularity or crystal size gradients, and/or stratified grain size sorting (Weiss et al., 2015).

4.4.2. Site Evaluation

Following the arguments regarding magnetization and timing in SO1 and orientation in SO2, NES is the preferred site, followed by CH and JE.

4.5. SO5 — Magnetic Carriers 4.5.1. Science Objective

Many studies have addressed the question of what mineral carriers could produce the observed strong Martian crustal magnetic fields (e.g., Arkani-Hamed, 2005; Dunlop & Arkani-Hamed, 2005; Kletetschka et al., 2000). The most discussed candidates are single-domain (SD) and multidomain (MD) magnetite, MD hematite, pyrrhotite, or ilmenite lamellae. Of these, SD magnetite can acquire strong TRM, and its relatively high Curie temperature of 580° C allows for up to 40–50 km of magnetized crust given the present Martian geotherm. Additionally, it has been argued that fine-grained SD magnetite could preserve its remanence over 4 Gyr (Sato et al., 2018).

There are a number of magnetically interesting Martian meteorites whose magnetic properties have been described and are dominated by magnetite: SD-sized magnetite has been discovered in the Amazonian-aged nakhlites and shergottites (e.g., Cisowski, 1986, Funaki et al., 2009, Jambon et al., 2016, Herd et al., 2017). However, the magnetic properties of these younger rocks may differ from those of older, more strongly magnetized

crust. SD magnetite is difficult to form during slow intrusive cooling events (Butler, 1992), and many alternative formation processes (e.g., Quesnel et al., 2009, Gunnlaugsson et al., 2006, Scott & Fuller, 2004) have been suggested. MD hematite is another good candidate, mainly due to its efficient TRM (Dunlop & Kletetschka, 2001; Kletetschka et al., 2000), and it has been detected in nakhlites (McCubbin et al., 2009) and NWA 7034 (Wittmann et al., 2015). In contrast, shergottite meteorites show that pyrrhotite is generally the main magnetic carrier, and for nakhlites it is titanomagnetite (Rochette et al., 2001). Although pyrrhotite can hold a strong TRM, its Curie temperature is only 320° C (Dunlop & Arkani-Hamed, 2005), thus limiting the depth of the magnetized layer to ~20 km, and it can also easily be demagnetized by impact shock (Rochette et al., 2003). NWA 7034 is dominated by magnetite and contains maghemite that could explain the observed strong crustal fields on Mars, and hydrothermal alteration may have led to its mineralogy (Gattacceca et al., 2014). The oldest Martian meteorite, ALH 84001, contains SD magnetite (Antretter et al., 2003; Weiss et al., 2002) and pyrrhotite (Weiss et al., 2008).

Because of the overall basaltic composition of most Martian crustal rocks (McLennan & Grotzinger, 2008), reflecting the scarcity of clastic felsic sediments and chemical weathering, ferromagnetic minerals are abundant on the surface (McSween et al., 2009). This is true even for Martian sediments like sandstones and claystones, which are far more enriched in ferromagnetic minerals than their typical counterparts on Earth. For example, magnetite, hematite, and pyrrhotite are each present at 1-5 wt.% levels in mudstones and aeolian deposits in Gale crater (Vaniman et al., 2013), and hematite is present at 10 wt.% in the sulfate-rich outcrops of Meridiani Planum (Christensen et al., 2004). Pyrrhotite has also been found at Gale crater (Vaniman et al., 2013). Even Martian airborne dust is highly magnetic (~0.3-0.5 wt.% magnetite; Goetz et al. 2008). These pyrrhotite and magnetite abundances are broadly similar to those of Martian meteorites. Ilmenite lamellae have also been suggested as a candidate mineralogy with properties allowing for high NRM and coercivity and requiring a layer as thin as ~10 km to explain the observed strong magnetic anomalies (McEnroe et al., 2004).

There are still many open questions regarding magnetic carriers on Mars. In particular, it is not sufficiently understood how the observed strong magnetization of Mars may be explained given the most common ferrimagnetic or ferromagnetic minerals and a magnetizing field strength similar to Earth's core dynamo (Morschhauser et al., 2018). Furthermore, knowing the magnetic mineralogy would allow understanding of how impacts, volcanic eruptions, and viscous decay influence NRM. These would, in turn, better constrain the core dynamo timing and magnetization depths. Finally, the nature and abundances of ferromagnetic minerals may place fundamental constraints on how the crust has been aqueously altered (Scott & Fuller, 2004), which has major implications for habitability (Grotzinger et al., 2014). Returned samples from Mars could greatly help answering these questions, in particular if they are selected from igneous rocks that crystallized in the deeper crust.

4.5.2. Site Evaluation

The investigation of magnetic carriers does not necessarily require magnetized rock but minerals that could carry magnetization when exposed to a magnetic field. A variety of different samples would offer a chance to explore a variety of mineralogies. We give a slight preference to CH, as we do know that rocks carry magnetization and we will therefore find possible carriers. However, the JE and NES landing sites are located closely to the Syrtis volcanic plains and may therefore also be suitable landing sites in this respect.

4.6. SO6—Sediment Sourcing and Fluid Flow

4.6.1. Science Objective

Detrital remanent magnetization is a further mechanism to magnetize rocks. The sample acquires magnetization by alignment with the ambient field during or past deposition. Fluid flow can lead to alignment of ferromagnetic particles when they are exposed to a magnetic field (Butler, 1992). Vice versa, if such magnetic anisotropy was detected, we can inversely infer paleo fluid flow direction. Also, alteration of ferromagnetic minerals (such as hematite or maghemite) would suggest extensive exposure to liquid water, whereas less oxidized magnetic minerals (magnetite) would suggest transient water. Requirements for a landing site should include signs of fluid flow ideally during an active global dynamo field or in regions of strong crustal magnetization.

4.6.2. Site Evaluation

SO6 follows the same argument as SO5, where all sites fulfill the morphological requirement of fluid flow at the surface, but CH also carries magnetization.

5. Discussion: Paleomagnetic Considerations for Mars 2020 Final Site Selection

From a satellite data perspective, the obvious first choice for a landing site would be CH, since it is the only site with considerable magnetization as seen from orbit. However, SO1, SO2, and SO4 require sampled material to be magnetized by a dynamo field, and the best likelihood of this is if the samples are pre-Noachian or Early Noachian. Material around CH is believed to be Late Noachian and younger. A sample from this site therefore has a chance of not addressing the science objectives related to an ancient paleodynamo field. Magnetization inferred from orbital data might also reflect magnetized material underlying the surface volcanic unit. Lava cooling in the magnetized crater could then still carry an NRM (even in absence of a dynamo field) but one that was acquired in presence of nearby crustal fields resulting from the magnetized basement units. This would allow objectives that do not require Early Noachian material, for example, SO3, SO5, and SO6, to still be addressed. Furthermore, if, as suggested by some studies (Langlais & Purucker, 2007; Milbury et al., 2012; Schubert et al., 2000), the dynamo extended into and beyond the Late Noachian, material from CH would allow S01 and, depending on geological context, possibly S02 to be addressed.

Megabreccia blocks have been observed in central peaks and rims of Noachian impact basins; specifically eight have been mapped within the NES landing site (and JE for an extended mission), which make this site an attractive choice (Weiss et al., 2018). Because many megabreccia blocks are layered and adjacent blocks often originate from similar original bedrock locations, they might provide access to hundreds of meters of ancient crust (Weiss et al., 2018). The chance of sampling pre-Noachian and Noachian material as well as intact stratigraphy would address SO1, SO2, and SO4. However, the best exposures of megabreccia deposits are currently outside the suggested NES landing ellipse and a slight change in landing location was proposed to accommodate access to them (Weiss et al., 2018). We note that the absence of regional magnetization inferred from orbital measurements is still compatible with the presence of strong localized magnetizations. By analogy, lunar surface magnetic fields in the vicinity of the Apollo 12 landing site, inferred from orbital data, are very weak, but specific samples from this landing site have shown evidence for magnetizations acquired in a dynamo field (e.g., Cournède et al., 2012).

Selection of magnetized samples would be greatly aided by having a magnetometer as part of the scientific instrument payload. This would allow outcrops to be surveyed prior to sampling, as well as permit a ground-based survey of the crustal magnetic signature (Diáz-Michelena et al., 2016).

The overall location of NES and JE might be a disadvantage because they are adjacent to the large impact basin Isidis. The 1,352-km-diameter basin is dated at 3.85–4.06 Ga (Frey, 2008), and the absence of magnetization within the basin has been interpreted as a demagnetization signal. Impacts can also result in stratigraphically inverted sections, with no easy way to identify the inversion. This could negatively impact achieving objective SO4. Shock and thermal alteration of nearby rocks may be an issue, and evaluation of these effects will be important for SO1, SO3, and SO5. In contrast, Gusev crater in CH is of Early Noachian age, and so the crater basement acquired magnetization after the impact with no subsequent impact-related disturbance of possible samples.

In conclusion, NES provides access to megabreccia, the oldest known materials on Mars and the basement unit of NES likely also contains Early Noachian materials. Thus, this site is the top choice for paleomagnetic sampling. The other sites also offer promising opportunities: CH is a good choice, motivated by the strong crustal fields measured from orbit and the absence of a major postdynamo impact crater. It may not be possible to address questions related to an ancient (pre-4.1 Ga) dynamo field with samples from CH, but it should be possible to test whether a dynamo was present during the Late Noachian or Early Hesperian period. Furthermore, the mineral grains and clasts in sediments around CH could be as old as Early Noachian because they are likely derived from Ma'adim Valley. Similar possibilities exist at the JE landing site, where Late Noachian-Early Hesperian material is abundant, sediments could be derived from megabreccia in the JE watershed, and where an extended mission could access such watershed rocks directly.

6. Conclusion

Martian return samples will revolutionize understanding of our neighbor planet, including from a magnetics perspective. All of the remaining three candidate landing sites will provide unprecedented information about the history and record of the magnetic field, and we suggest that the preferred site is NES. Satellite observations indicate weak or no magnetic field signatures at NES and JE, and a moderate magnetic field signature at CH. However, to address the magnetic signature generated by the ancient dynamo field, samples of pre-Noachian and Early Noachian age are required, which is likely not the case for CH and JE, but is the case for NES. Specifically, the megabreccia deposits may provide samples, that represent earliest records of the Martian paleodynamo field and also provide insight into paleoclimate and habitability including aqueous processes, biosignatures, planetary accretion, and igneous differentiation (Weiss et al., 2018). A recommendation for Mars 2020 or future lander sample missions is the inclusion of a magnetometer to survey potential outcrops for magnetic field signatures ensuring the best possible target sample and increased science return for the overall mission.

References

Acuna, M. H., Connerney, J. E. P., Ness, N. F, Lin, R. P., Mitchell, D., Carlson, C. W., et al. (1999). Global distribution of crustal magnetization discovered by the Mars Global Surveyor MAG/ER experiment. *Science*, 284(5415), 790–793.

Amit, H., Christensen, U. R., & Langlais, B. (2011). The influence of degree-1 mantle heterogeneity on the past dynamo of Mars. *Physics of the Earth and Planetary Interiors*, 189(1-2), 63–79. https://doi.org/10.1016/j.pepi.2011.07.008

Andrews-Hanna, J. C., Zuber, M. T., & Banerdt, W. B. (2008). The Borealis basin and the origin of the martian crustal dichotomy. *Nature*, 453(7199), 1212–1215. http://www.nature.com/doifinder/10.1038/nature07011

Antretter, M., Fuller, M., Scott, E., Jackson, M., Moskowitz, B., & Solheid, P. (2003). Paleomagnetic record of Martian meteorite ALH84001. Journal of Geophysical Research, 108(E6), 5049. https://doi.org/10.1029/2002JE001979

Arkani-Hamed, J. (2004). Timing of the Martian core dynamo. *Journal of Geophysical Research*, 109, E03006. https://doi.org/10.1029/2005JE002535

Arkani-Hamed, J. (2005). On the possibility of single-domain/pseudo-single-domain magnetic particles existing in the lower crust of Mars: Source of the strong magnetic anomalies. *Journal of Geophysical Research*, *110*, E12009. https://doi.org/10.1029/2005JE002535 Aster, R. C., Borchers, B., & Thurber, C. H. (2011). *Parameter Estimation and Inverse Problems*. Amsterdam: Elsevier Academic Press.

Banerdt, Boschi, L., Christensen, U., Dehant, V., Giardini, D., Goetz, W., et al. (2013). InSight: A discovery mission to explore the interior of Mars. In *Lunar and Planetary Science Conference* (1915 pp.). http://adsabs.harvard.edu/abs/2013LPI....44.1915B

Blakely, R. J. (1995). Potential theory in gravity and magnetic applications: Cambridge University Press.

 Bramble, M. S., Mustard, J. F., & Salvatore, M. R. (2017). The geological history of Northeast Syrtis Major, Mars. *Icarus*, 293, 66–93.
 Butler, R. F. (1992). *Paleomagnetism: Magnetic domains to geologic terranes* (Vol. 319). Boston, MA: Blackwell Scientific Publications.
 Cain, J. C. (2003). An n = 90 internal potential function of the Martian crustal magnetic field. *Journal of Geophysical Research*, 108, 5008. https://doi.org.10.1029/2000JE001487

 Christensen, P. R., Wyatt, M. B., Glotch, T. D., Rogers, A. D., Anwar, S., Arvidson, R. E., et al. (2004). Mineralogy at Meridiani Planum from the Mini-TES Experiment on the Opportunity Rover. *Science*, *306*(5702), 1733–1739. http://science.sciencemag.org/content/306/5702/1733
 Cisowski, S. M. (1986). Magnetic studies on Shergotty and other SNC meteorites. *Geochimica et Cosmochimica Acta*, *50*, 1043–1048.
 Collinson, D. W. (1997). Magnetic properties of Martian meteorites: Implications for an ancient Martian magnetic field. *Meteoritics and*

Planetary Science, 32, 803–811.

Connerney, J. E. P., Acuna, M. H., Ness, N. F., Kletetschka, G., Mitchell, D. L., Lin, R. P., & Reme, H. (2005). Tectonic implications of Mars crustal magnetism. *Proceedings of the National Academy of Sciences*, 102(42), 14970–14975. http://www.pnas.org/cgi/doi/10.1073/pnas.0507469102

Connerney, J. E. P., Acuna, M. H., Wasilewski, P. J., Ness, N. F., Reme, H., Mazelle, C., et al. (1999). Magnetic lineations in the ancient crust of Mars. Science, 284(5415), 794–798.

Cournède, C., Gattacceca, J., & Rochette, P. (2012). Magnetic study of large Apollo samples: Possible evidence for an ancient centered dipolar field on the Moon. *Earth and Planetary Science Letters*, 331-332, 31–42. https://doi.org/10.1016/j.epsl.2012.03.004

Cox, A., Doell, R. R., & Dalrymple, G. B. (1964). Reversals of the Earth's magnetic field. Science, 144(2), 1537–1543.

Dehant, V., Lammer, H., Kulikov, Y. N., Grießmeier, J. M., Breuer, D., Verhoeven, O., et al. (2007). Planetary magnetic dynamo effect on atmospheric protection of early Earth and Mars. *Space Science Reviews*, 129(1-3), 279–300.

Diáz-Michelena, M., Kilian, R., Sanz, R., Rios, F., & Baeza, O. (2016). Mars MOURA magnetometer demonstration for high-resolution mapping on terrestrial analogues. *Geoscientific Instrumentation, Methods and Data Systems*, 5(1), 127–142.

Driscoll, P., & Olson, P. (2009). Effects of buoyancy and rotation on the polarity reversal frequency of gravitationally driven numerical dynamos. *Geophysical Journal International*, 178(3), 1337–1350.

Dunlop, D. J., & Arkani-Hamed, J. (2005). Magnetic minerals in the Martian crust. *Journal of Geophysical Research*, 110, E12S04. https://doi.org.10.1029/2005JE002404

Dunlop, D. J., & Kletetschka, G. (2001). Multidomain hematite: A source of planetary magnetic anomalies? *Geophysical Research Letters*, 28, 3345–3348.

Ehlmann, B. L., Anderson, F. S., Andrews-Hanna, J., Catling, D. C., Christensen, P. R., Cohen, B. A., et al. (2016). The sustainability of habitability on terrestrial planets: Insights, questions, and needed measurements from Mars for understanding the evolution of Earth-like worlds. *Journal of Geophysical Research : Planets, 121*, 1927–1961. https://doi.org/10.1002/2016JE005134

Ehlmann, B. L., & Mustard, J. F. (2012). An in-situ record of major environmental transitions on early Mars at Northeast Syrtis Major. *Geophysical Research Letters*, 39, L11202. https://doi.org/10.1029/2012GL051594

Fairén, A. G., Ruiz, J., & Anguita, F. (2002). An origin for the linear magnetic anomalies on Mars through accretion of terranes: Implications for dynamo timing. *Icarus*, 160(1), 220–223.

Fassett, C. I., & Head, J. W. (2011). Sequence and timing of conditions on early Mars. *Icarus*, 211(2), 1204–1214. https://doi.org/10.1016/j.icarus.2010.11.014

Filiberto, J., & Schwenzer, S. P. (2013). Alteration mineralogy of Home Plate and Columbia Hills-Formation conditions in context to impact, volcanism, and fluvial activity. *Meteoritics and Planetary Science*, 48(10), 1937–1957.

Frey, H. (2008). Ages of very large impact basins on Mars: Implications for the late heavy bombardment in the inner solar system. *Geophysical Research Letters*, 35, L13203. https://doi.org/10.1029/2008GL033515

Funaki, M., Hoffmann, V., & Imae, N. (2009). Estimate of the magnetic field of Mars based on the magnetic characteristics of the Yamato 000593 nakhlite. *Meteoritics and Planetary Science*, 44, 1179–1191.

Gattacceca, J., & Rochette, P. (2004). Toward a robust normalized magnetic paleointensity method applied to meteorites. *Earth and Planetary Science Letters*, 227(3-4), 377–393.

Acknowledgments

We acknowledge support from the University of British Columbia through a Four Year Fellowship (A. M.), the Natural Sciences and Engineering Research Council of Canada, the Canadian Space Agency and the InSight Mission (C. L. J.). the Deutsche Forschungsgemeinschaft within the Priority 1788 'Dynamic Earth' under the grant LE2477/7-1 (F. V.), CNES via the InSight project, and by the European Union's Horizon 2020 research and innovation programme under grant agreement 730041 (B. L.). The MGS and MAVEN magnetic field data used in this study are archived in the Planetary Data System.

Gattacceca, J., Rochette, P., Scorzelli, R. B., Munayco, P., Agee, C., Quesnel, Y., et al. (2014). Martian meteorites and Martian magnetic anomalies: A new perspective from NWA 7034. *Geophysical Research Letters*, 41, 4859–4864. https://doi.org/10.1002/2014GL060464 Goetz, W., Hviid, S. F., Kinch, K. M., & Madsen, M. B. (2008). Magnetic properties results from surface landers and rovers. In J. F. Bell III

(Ed.), The Martian Surface: Composition, Mineralogy and Physical Properties (pp. 366–380). New York: Cambridge University Press. https://doi.org/10.1017/CBO9780511536076.017

Grotzinger, J. P., Sumner, D. Y., Kah, L. C., Stack, K., Gupta, S., Edgar, L., et al. (2014). A habitable fluvio-lacustrine environment at Yellowknife Bay, Gale Crater, Mars. *Science*, 343(6169). http://science.sciencemag.org/content/343/6169/1242777

Gubbins, D., Ivers, D., Masterton, S. M., & Winch, D. E. (2011). Geophysical Journal International. *Geophysical Journal International*, 187(1), 99–117.

Gunnlaugsson, H. P., Helgason, Ö., Kristjánsson, L., Nørnberg, P., Rasmussen, H., Steinþórsson, S., & Weyer, G. (2006). Magnetic properties of olivine basalt: Application to Mars. *Physics of the Earth and Planetary Interiors*, *154*, 276–289.

Harrison, K. H., & Grimm, R. E. (2002). Controls on Martian hydrothermal systems: Application to valley network and magnetic anomaly formation. Journal of Geophysical Research, 107(E5), 5025. https://doi.org/5025,10.1029/2001JE001616

Herd, C. D. K., Walton, E. L., Agee, C. B., Muttik, N., Ziegler, K., Shearer, C. K., et al. (2017). The Northwest Africa 8159 Martian meteorite: Expanding the martian sample suite to the early Amazonian. *Geochimica et Cosmochimica Acta*, 218, 1–26.

Hiesinger, H., & Head, J. W. (2004). The Syrtis Major volcanic province, Mars: Synthesis from Mars Global Surveyor data. *Journal of Geophysical Research*, *109*, E01004. http://doi.wiley.com/10.1029/2003JE002143

Hood, L. L., & Zakharian, A. (2001). Mapping and modeling of magnetic anomalies in the northern polar region of Mars. Journal of Geophysical Research, 106(E7), 14,601–14,619. https://doi.org/10.1029/2000JE001304

Jakosky, B. M., & Phillips, R. J. (2001). Mars' volatile and climate history. Nature, 412(6843), 237-244.

Jambon, A., Sautter, V., Barrat, J. A., Gattacceca, J., Rochette, P., Boudouma, O., et al. (2016). Northwest Africa 5790: Revisiting nakhlite petrogenesis. *Geochimica et Cosmochimica Acta*, 190(January), 191–212. https://doi.org/10.1016/j.gca.2016.06.032

Johnson, B. C., Blair, D. M., Collins, G. S., Melosh, H. J., Freed, A. M., Taylor, G. J., et al. (2016). Formation of the Orientale lunar multiring basin. Science, 354(6311), 441–444. http://science.sciencemag.org/content/354/6311/441

Kletetschka, G., Wasilewski, P. J., & Taylor, P. T. (2000). Hematite vs. magnetite as the signature for planetary magnetic anomalies? Physics of the Earth and Planetary Interiors, 119, 259–267.

Kletetschka, G., Wasilewski, P. J., & Taylor, P. T. (2000). Mineralogy of the sources for magnetic anomalies on Mars. *Meteoritics and Planetary Science*, 35(5), 895–899.

Kuang, W., Jiang, W., & Wang, T. (2008). Sudden termination of Martian dynamo?: Implications from subcritical dynamo simulations. Geophysical Research Letters, 35, L14204. https://doi.org/10.1029/2008GL034183

Langlais, B., & Purucker, M. (2007). A polar magnetic paleopole associated with Apollinaris Patera, Mars. *Planetary and Space Science*, 55(3), 270–279.

Langlais, B., Purucker, M., & Mandea, M. (2004). Crustal magnetic field of Mars. Journal of Geophysical Research, 109, E02008. https://doi.org/10.1029/2003JE002048

Langlais, B., Thébault, E., & Houliez, A. (2017). The Martian crustal magnetic field: A global view as seen from MGS and MAVEN measurements. In *European Planetary Science Congress*, (vol. 11)

Lillis, R. J., Frey, H. V., & Manga, M. (2008). Rapid decrease in Martian crustal magnetization in the Noachian era: Implications for the dynamo and climate of early Mars. *Geophysical Research Letters*, 35, L14203. https://doi.org/10.1029/2008GL034338

Lillis, R. J., Manga, M., Mitchell, D. L., Lin, R. P., & Acuna, M. H. (2006). Unusual magnetic signature of the Hadriaca Patera Volcano: Implications for early Mars. *Geophysical Research Letters*, 33, L03202. https://doi.org/10.1029/2005GL024905

- Lillis, R. J., Robbins, S., Manga, M., Halekas, J. S., & Frey, H. V. (2013). Time history of the Martian dynamo from crater magnetic field analysis. Journal of Geophysical Research: Planets Planets, 118, 1488–1511. https://doi.org/10.1002/jgre.20105
- Marinova, M. M, Aharonson, O., & Asphaug, E. I. (2008). Mega-impact formation of the Mars hemispheric dichotomy. *Nature*, 453(7199), 1216–1219.

Maus, S., & Haak, V. (2003). Magnetic field annihilators: Invisible magnetization at the magnetic equator. *Geophysical Journal International*, 155(2), 509–513.

Mayhew, M. A. (1979). Inversion of satellite magnetic anomaly data. Journal of Geophysics Zeitschrift Geophysik, 45, 119–128.

McCoy, T. J., Sims, M., Schmidt, M. E., Edwards, L., Tornabene, L. L., Crumpler, L. S., et al. (2008). Structure, stratigraphy, and origin of Husband Hill, Columbia Hills, Gusev Crater, Mars. Journal of Geophysical Research, 113, E06503. https://doi.org/10.1029/2007JE003041

McCubbin, F. M., Tosca, N. J., Smirnov, A., Nekvasil, H., Steele, A., Fries, M., et al. (2009). Hydrothermal jarosite and hematite in a pyroxene-hosted melt inclusion in martian meteorite Miller Range (MIL) 03346: Implications for magmatic-hydrothermal fluids on Mars.

Geochimica et Cosmochimica Acta, 73(16), 4907–4917. https://doi.org/10.1016/j.gca.2009.05.031
McEnroe, S. A., Brown, L. L., & Robinson, P. (2004). Earth analog for Martian magnetic anomalies: Remanence properties of hemo-ilmenite norites in the Bjerkreim-Sokndal intrusion, Rogaland, Norway. *Journal of Applied Geophysics*, 56(3), 195–212.

McKenzie, D. (1999). Plate tectonics on Mars? Nature, 399(6734), 307-308.

McLennan, S. M., & Grotzinger, J. P. (2008). The sedimentary rock cycle of Mars. In J. F. Bell (Ed.), The Martian Surface—Composition, Mineralogy, and Physical Properties (pp. 541–577). Cambridge: Cambridge University Press.

McSween, H. Y., Taylor, G. J., & Wyatt, M. B. (2009). Elemental composition of the Martian crust. Science, 324(5928), 736–739.

Merrill, R. T., & McElhinny, M. W. (1983). The Earth's Magnetic Field: Its History, Origin and Planetary Perspective. London: Academic Press. Metelkin, D. V., Vernikovsky, V. A., Kazansky, A. Y., & Wingate, M. T. D. (2010). Late Mesozoic tectonics of Central Asia based on paleomagnetic evidence. Gondwana Research, 18(2-3), 400–419. https://doi.org/10.1016/j.gr.2009.12.008

Milbury, C., Schubert, G., Raymond, C. A., Smrekar, S. E., & Langlais, B. (2012). The history of Mars' dynamo as revealed by modeling magnetic anomalies near Tyrrhenus Mons and Syrtis Major. *Journal of Geophysical Research*, *117*, E10007. https://doi.org/10.1029/2012JE004099

Mittelholz, A., Johnson, C. L., & Morschhauser, A. (2018). A new magnetic field activity proxy for Mars from MAVEN data. *Geophysical Research Letters*, 45(12), 5899–5907. https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2018GL078425

Mittelholz, A., Morschhauser, A., & Johnson, C. (2017). New crustal field models of Mars. In AGU Fall Meeting Abstracts.

Monteux, J., Amit, H., Choblet, G., Langlais, B., & Tobie, G. (2015). Giant impacts, heterogeneous mantle heating and a past hemispheric dynamo on Mars. *Physics of the Earth and Planetary Interiors, 240*, 114–124. https://doi.org/10.1016/j.pepi.2014.12.005

Morris, R. V., Ruff, S. W., Gellert, R., Ming, D. W., Arvidson, R. E., Clark, B. C., et al. (2010). Identification of carbonate-rich outcrops on Mars by the Spirit rover. *Science*, 329(5990), 421–424. http://science.sciencemag.org/content/329/5990/421

Morschhauser, A., Lesur, V., & Grott, M. (2014). A spherical harmonic model of the lithospheric magnetic field of Mars. *Journal of Geophysical Research: Planets, 119*, 1162–1188. https://doi.org/10.1002/2013JE004555

Morschhauser, A., Vervelidou, F., Thomas, P., Grott, M., Lesur, V., & Gilder, S. A. (2018). Mars' crustal magnetic field. In H. Lühr, J. Wicht, S. A. Gilder, & M. Holschneider (Eds.), *Magnetic Fields in the Solar System : Planets, Moons and Solar Wind In teractions* (pp. 331–356). Cham: Springer International Publishing. https://doi.org/10.1007/978-3-319-64292-5-12

Mustard, J, Adler, M, Allwood, A, Bass, D, Beaty, D, Bell, J, et al. (2013). Report of the Mars 2020 science definition team. *Mars Exploration Program Analysis Group (MEPAG)*, *Cl*, 155–205. http://mepag.jpl.nasa.gov/reports/MEP/Mars_2020_SDT_Report_Appendix.pdf Nimmo, F., & Tanaka, K. (2005). Early crustal evolution of Mars. *Annual Review of Earth and Planetary Sciences*, 33(1), 133–161.

https://doi.org/10.1146/annurev.earth.33.092203.122637

Olson, P., Deguen, R., Hinnov, L. A., & Zhong, S. (2013). Controls on geomagnetic reversals and core evolution by mantle convection in the Phanerozoic. *Physics of the Earth and Planetary Interiors, 214*, 87–103. https://doi.org/10.1016/j.pepi.2012.10.003

Pétrélis, F., Besse, J., & Valet, J. P. (2011). Plate tectonics may control geomagnetic reversal frequency. *Geophysical Research Letters*, 38, L19303. https://doi.org/10.1029/2011GL048784

Parker, R. L. (2003). Ideal bodies for Mars magnetics. Journal of Geophysical Research, 108(E1), 5006. https://doi.org/10.1029/2001JE001760Purucker, M., Ravat, D., Frey, H., Voorhies, C., Sabaka, T., & Acuna, M. (2000). An altitude-normalized magnetic map of Mars and its interpretation. Geophysical Research Letters, 27(16), 2449–2452.

Quesnel, Y., Sotin, C., Langlais, B., Costin, S., Mandea, M., Gottschalk, M., et al. (2009). Serpentinization of the martian crust during Noachian. Earth and Planetary Science Letters, 277(1-2), 184–193. http://linkinghub.elsevier.com/retrieve/pii/S0012821X0800678X

Rochette, P. (2006). Crustal magnetization of Mars controlled by lithology or cooling rate in a reversing dynamo? *Geophysical Research Letters*, 33, L02202. https://doi.org/10.1029/2005GL024280

Rochette, P., Fillion, G., Ballou, R., Brunet, F., Ouladdiaf, B., & Hood, L. (2003). High pressure magnetic transition in pyrrhotite and impact demagnetization on Mars. *Geophysical Research Letters*, 30(13), 1683. https://doi.org/10.1029/2003GL017359

Rochette, P., Gattacceca, J., Chevrier, V., Hoffmann, V., Lorand, J. P., Funaki, M., et al. (2005). Matching Martian crustal magnetization and magnetic properties of Martian meteorites. *Meteoritics and Planetary Science*, 40, 529.

Rochette, P., Gattacceca, J., Chevrier, V., Mathé, P. E., Menvielle, M., & Mappa Science Team (2006). Magnetism, iron minerals, and life on Mars. Astrobiology, 6(3), 423–436.

Rochette, P., Lorand, J. P., Fillion, G., & Sautter, V. (2001). Pyrrhotite and the remanent magnetization of SNC meteorites: A changing perspective on Martian magnetism. *Earth and Planetary Science Letters*, 190(1-2), 1–12.

Ruff, S. W., Farmer, J. D., Calvin, W. M., Herkenhoff, K. E., Johnson, J. R., Morris, R. V., et al. (2011). Characteristics, distribution, origin, and significance of opaline silica observed by the Spirit rover in Gusev crater, Mars. *Journal of Geophysical Research*, 116, E00F23. https://doi.org/10.1029/2010JE003767

Ruff, S. W., Niles, P. B., Alfano, F., & Clarke, A. B. (2014). Evidence for a Noachian-aged ephemeral lake in Gusev crater, Mars. Geology, 42(4), 359–362.

Runcorn, S. K. (1975). An ancient lunar magnetic dipole field. *Nature*, 253(5494), 701–703.

Ryan, D. A., & Sarson, G. R. (2007). Are geomagnetic field reversals controlled by turbulence within the Earth's core? *Geophysical Research Letters*, *34*, L02307. https://doi.org/10.1029/2006GL028291

Sato, M., Yamamoto, Y., Nishioka, T., Kodama, T., Mochizuki, N., Ushioda, M., et al. (2018). Constraints on the source of the Martian magnetic anomalies inferred from relaxation time of remanent magnetization. *Geophysical Research Letters*, 45, 6417–6427. https://doi.org/10.1029/2018GL077498

Schubert, G., Russell, C. T., & Moore, W. B. (2000). Geophysics: Timing of the Martian dynamo. *Nature*, 408(6813), 666–667. http://www.nature.com/doifinder/10.1038/35047163

Scott, E. R. D., & Fuller, M. (2004). A possible source for the Martian crustal magnetic field. *Earth and Planetary Science Letters*, 220, 83–90. Sleep, N. H. (1994). Martian plate tectonics. *Journal of Geophysical Research*, 99(E3), 5639–5655.

Smith, D. E., Zuber, M. T., Frey, H. V., Garvin, J. B., Muhleman, D. O., Pettengill, G. H., et al. (2001). Mars Orbiter Laser Altimeter: Experiment summary after the first year of global mapping of Mars. *Journal of Geophysical Research*, 106(2000), 689–722.

Sprenke, K. F., & Baker, L. L. (2000). Magnetization, paleomagnetic poles, and polar wander on Mars. Icarus, 147(1), 26-34.

Squyres, S. W., Arvidson, R. E., Ruff, S., Gellert, R., Morris, R. V., Ming, D. W., et al. (2008). Detection of silica-rich deposits on Mars. *Science*, 320(5879), 1063–1067. http://science.sciencemag.org/content/320/5879/1063

Stanley, S., Elkins-Tanton, L., Zuber, M. T, & Parmentier, E. M. (2008). Mars' paleomagnetic field as the result of a single-hemisphere dynamo. *Science*, 321(5897), 1822–1825. http://science.sciencemag.org/content/321/5897/1822

Stevenson, D. J. (2001). Mars' core and magnetism. *Nature*, *412*(6843), 214–219.

Takahashi, F., Tsunakawa, H., Shimizu, H., Shibuya, H., & Matsushima, M. (2014). Reorientation of the early lunar pole. *Nature Geoscience*, 7(6), 409–412.

Tanaka, K. L., Skinner, J. A., Dohm, J. M., Irwin, R. P., Kolb, E. J., Fortezzo, C. M., Platz, T., Michael, G. G., & Hare, T. M. (2014). Geologic map of Mars, pp. 3292: U.S. Geological Survey Geologic Investigations. http://pubs.usgs.gov/sim/3292/

Thomas, P., Grott, M., Morschhauser, A., & Vervelidou, F. (2018). Paleopole reconstruction of Martian magnetic field anomalies. *Journal of Geophysical Research: Planets*, 123, 1140–1155. https://doi.org/10.1002/2017JE005511

van Kan Parker, M., Zegers, T., Kneissl, T., Ivanov, B., Foing, B., & Neukum, G. (2010). 3D structure of the Gusev Crater region. Earth and Planetary Science Letters, 294(3-4), 411–423. https://doi.org/10.1016/j.epsl.2010.01.013

Vaniman, D. T., Bish, D. L., Ming, D. W., Bristow, T. F., Morris, R. V., Blake, D. F., et al. (2013). Mineralogy of a Mudstone at Yellowknife Bay, Gale Crater, Mars. Science, 343(1243480). http://science.sciencemag.org/content/early/2013/12/05/science.1243480

Vervelidou, F., Lesur, V., Grott, M., Morschhauser, A., & Lillis, R. J. (2017). Constraining the date of the Martian dynamo shutdown by means of crater magnetization signatures. *Journal of Geophysical Research: Planets*, 122, 2294–2311. https://doi.org/10.1093/gji/ggx400

Vervelidou, F., Lesur, V., Morschhauser, A., Grott, M., & Thomas, P. (2017). On the accuracy of palaeopole estimations from magnetic field measurements. *Geophysical Journal International*, 211(3), 1669–1678. http://academic.oup.com/gji/article/211/3/1669/4222794

Watters, T. R., McGovern, P. J., & Irwin, R. P. (2007). Hemispheres apart: The crustal dichotomy on Mars. Annual Review of Earth and Planetary Sciences, 35(1), 621–652. http://www.annualreviews.org/doi/10.1146/annurev.earth.35.031306.140220

Weil, A. B., Van der Voo, R., Mac Niocaill, C., & Meert, J. G. (1998). The proterozoic supercontinent Rodinia:

Paleomagnetically derived reconstructions for 1100 to 800 Ma. *Earth and Planetary Science Letters*, 154(1-4), 13–24. http://linkinghub.elsevier.com/retrieve/pii/S0012821\$\times\$97001271

Weiss, B. P., Beaty, D. W, McSween, H. Y., Carrier, B. L., Czaja, A. D., Goreva, Y. S., et al. (2018). Paleomagnetic studies of returned samples from Mars: 2nd International Mars Sample Return 2018.

Weiss, B. P., Beaty, D. W., McSween, H. Y. Jr., Czaja, A. D., Goreva, Y., Hausrath, E., et al. (2016). Planning for the paleomagnetic investigations of returned samples from Mars. Abstract presented at Fall Meeting, AGU, San Francisco, CA.

Weiss, B. P., Fong, L. E., Vali, H., Lima, E. A., & Baudenbacher, F. J. (2008). Paleointensity of the ancient Martian magnetic field. *Geophysical Research Letters*, 35, L23207. https://doi.org/10.1029/2008GL035585

Weiss, B. P., Vali, H., Baudenbacher, F. J., Kirschvink, J. L., Stewart, S. T., & Shuster, D. L. (2002). Records of an ancient Martian magnetic field in ALH84001. *Earth and Planetary Science Letters*, 201, 449–463.

Werner, S. C. (2008). The early martian evolution — Constraints from basin formation ages. *Icarus*, 195, 45–60.

Whaler, K. A., & Langel, R. A. (1996). Minimal crustal magnetizations from satellite data. *Physics of the Earth and Planetary Interiors*, 48(96), 303–319.

Whaler, K. A., & Purucker, M. E. (2005). A spatially continuous magnetization model for Mars. *Journal of Geophysical Research*, 110, E09001. https://doi.org/10.1029/2004JE002393

Wittmann, A., Korotev, R. L., Jolliff, B. L., Irving, A. J., Moser, D. E., Barker, I., & Rumble, D. (2015). Petrography and composition of Martian regolith breccia meteorite Northwest Africa 7475. *Meteoritics and Planetary Science*, *50*(2), 326–352.